## UNIVERSITY OF THE WESTERN CAPE

## **Faculty of Community and Health Sciences**

**Title:** Perceptions of pregnant women on reasons for late initiation of antenatal care in Nkwen Baptist Health Center, North West Region, Cameroon.

## **Denis Warri**

**MPH** 

Mini-Thesis submitted in partial fulfillment of the requirement of the Degree of Master of Public Health, in the School of Public Health, Faculty of Community and Health Sciences,

RIGHT STREET, STREET,

University of the Western Cape

UNI South Africa Y of the WESTERN CAPE

Supervisor: Prof Asha George

**August 2018** 

# **Contents**

Keywords	
Abbreviations	
Definition of TermsAbstract	
Background	
Methodology	
Results Conclusion and Recommendations	
Declaration	
Acknowledgements  Dedication	
CHAPTER 1: INTRODUCTION	
1.1 Study Background	
1.3 Purpose of the Study	
1.4 Outline of the Study	
CHAPTER 2: LITERATURE REVIEW	
2.1 Introduction	
2.2 Antenatal care and maternal health	
2.3 Initiation of first Antenatal Care visit      2.4 Factors associated with late initiation of Antenatal Care	
2.5 Conceptual Framework	
2.5.1 Core assumptions of the health belief model	
2.5.2 Perceived susceptibility related to early antenatal booking	
2.5.4 Perceived benefits of early initiation of antenatal care	
2.5.5 Perceived barriers contributing to late initiation of antenatal care	
2.5.6 Cues to action	9
2.5.8 Integration of factors associated with late initiation of antenatal care into the	onceptual
framework	9
2.6 Summary	11
CHAPTER 3: METHODOLOGY	12
3.1 Introduction	
3.3 Objectives of the Study	
3.4 Study Design	
3.5 Research Setting	
3.6 Sampling	
3.7 Data Collection	15
3.7.1 Data Collection Method	15
3.8 Data Management	16
3.9 Data Analysis	
3.9.1 Phase 1: Familiarization	16
3.9.2 Phase 2: Coding	
3.9.3 Phase 3: Identifying Themes	

3.9.4 Phase 4: Reviewing and Refining Themes	
3.10 Rigor	
CHAPTER 4: FINDINGS	19
4.1 Introduction	
4.2.1 Perceived susceptibility/perceived severity of antenatal care	20
4.2.1.1 Value of early antenatal care	
4.2.1.2 Pregnancy as a normal health condition	21
4.2.1.3 Ideal booking time	22
4.2.1.4 Obstetric history	22
4.2.2 Perceived barriers to antenatal care	23
4.2.2.1 Accessibility of antenatal care services	23
4.2.2.2 Cost of initiating antenatal care	24
4.2.2.3 Distance to health facility	24
4.2.3 Cues to action	25
4.2.3.1 Community health education	25
4.2.4 Self – Efficacy	25
4.2.4.1 Pregnancy disclosure	25
4.2.1.2 Support from spouse	27
4.2.1.3 Reaction from parents	27
4.2.1.3 Reaction from parents	28
5.1 Perceived susceptibility/ Perceived Severity due to late antenatal care initiation	
5.2 Perceived health systems barriers to early antenatal care	
5.4 Self – efficacy	
5.5 Study Limitations	31
CHAPTER 6: CONCLUSION AND RECOMMENDATIONS	31
6.1 Conclusion	
6.2 Recommendations	
7.0 REFERENCE LIST	34
8.0 APPENDICES	
Appendix 1. Participant Information Sheet	
Appendix 3. Interview Guide	
Appendix 4. UWC Ethics Approval	46
Appendix 5. CBC Health Services Approval Letter	
	<del> </del>

## **Keywords**

- Antenatal Care
- Maternal Mortality
- Pregnant Women
- Midwives
- Care-seeking
- Maternal Health
- Health Education
- Health Center
- North West Region
- Cameroon

## **Abbreviations**

ANC Antenatal Care

HIV Human Immunodeficiency Virus

IEC Information Education and Communication

IRB Institutional Review Board

KI Key Informant IVERSITY of the

P Participant CAPE

PMTCT Prevention of Mother to Child Transmission

TCA Thematic Coding Analysis

UNICEF United Nations Children Fund

USAID United States Agency for International Development

UWC University of the Western Cape

WHO World Health Organization

## **Definition of Terms**

**Antenatal care**: The care provided by skilled healthcare professionals to pregnant women and adolescent girls in order to ensure the best health conditions for both mother and baby during pregnancy (WHO, 2016).

**Late initiation of antenatal care:** When pregnant women make first appearance at an antenatal clinic after 12 weeks of gestation (WHO, 2016).

**Perceptions:** Impressions or interpretations based on the understanding of something or an experience.



#### **Abstract**

## **Background**

Antenatal care serves as a key entry point for a pregnant woman to receive a broad range of services and should be initiated at the onset of pregnancy (WHO, 2016). Cameroon has one of the highest maternal mortality ratios in the world (UNICEF, 2016). The majority of pregnant women in Cameroon initiate antenatal care after the first trimester (Njim, 2016). Most studies on initiation of antenatal care in Cameroon have not explored in greater depth the reasons why most of the pregnant women initiate antenatal care late.

## Methodology

The aim of the study is to understand the reasons why pregnant women initiate antenatal care late in Nkwen Baptist Health Center, North West Region, Cameroon. It is an exploratory study and applied purposive sampling to recruit eighteen pregnant women and three key informants for data collection through individual interviews. Pregnant women who initiated antenatal care after the first trimester were recruited during antenatal care clinics and interviewed in a room at the antenatal care unit. Key informants were midwives working at the antennal care unit. Participation in the study was voluntary. Participants were explained the purpose of the study and signed a consent form if they were willing to participate in the research. Participation in the research did not inhibit the respondent's access to care. Data was collected using an audio tape and analyzed using Thematic Coding Analysis (TCA) to identify recurring themes that emerged from the data to adequately describe the perceptions of respondents on the reasons for late initiation of antenatal care.

#### **Results**

Interviews of pregnant women and service providers revealed that pregnant women place low value on early antenatal care due to the fact that they perceive pregnancy as a normal health condition or not a serious issue that requires seeking health care. Furthermore previous pregnancy outcomes that were positive regardless of accessing care made them less motivated to initiate antenatal care early. Some of the participants who had unplanned pregnancies sought to delay initiating antenatal care because they planned to terminate the pregnancy. The booking system is perceived as user-unfriendly with overcrowded conditions, long waiting times and rudeness of some service providers. Cost of services and distance to health facilities with uncomfortable transport and poor road network were identified as perceived barriers. The absence of community health programmes, perceived lack of support

from parents and spouses, fear of bewitchment and stigma due to cultural beliefs about early initiation of antenatal care were also identified as variables influencing late initiation.

## **Conclusion and Recommendations**

Pregnant women lack information on the purpose of early antenatal care. Health systems barriers as well as socio cultural beliefs also have significant influence on timing of antenatal care initiation. Quality improvement by service providers and implementation of community health education programmes could improve timing of care seeking for antenatal care and thereby improve the maternal health status of women.



## **Declaration**

I, Denis Warri declare that the research study "Perceptions of pregnant women on reasons for late initiation of antenatal care in Nkwen Baptist Health Center, North West Region, Cameroon" is my work and has not been submitted before for any degree or examination at any other university, and that all sources I have used or quoted have been indicated and acknowledged by means of complete references.

Denis Warri

Signature:

Date: August 2018

## Acknowledgements

I wish to acknowledge the support, guidance and encouragement through out this study from my supervisor, Prof. Asha George. Her technical expertise and constructive advice contributed significantly in realising this project.

Special appreciation to the study participants for their willingness to participate in this study. The participants sacrificed their time and were open to share their experiences. The study would not have been possible without their participation.

I would like to acknowledge my employer, the Director of the Cameroon Baptist Convention Health Services and the management and staff of Nkwen Baptist Health Center for the opportunity and enabling environment to undertake this study.

Special thanks to my wife, Joyce and children; Chelsy, Carlson, Colby and Charnelle for their sacrifice and moral support through out the research process.



## **Dedication**

This thesis is dedicated to the Almighty God for His guidance and leadership.

This work is also dedicated to my wife, Joyce and children; Chelsy, Carlson, Colby and Charnelle as a source of inspiration to them as they grow to achieve their career goals.



#### **CHAPTER 1: INTRODUCTION**

This chapter provides background information on initiation of antenatal care among pregnant women, the problem statement, purpose and outline of the thesis.

## 1.1 Study Background

In 2015, about 303,000 women died from pregnancy related causes globally, with 99% of all maternal deaths occurring in low and middle income countries (WHO, 2018). Within low and middle income countries, the highest maternal mortality rates are found in sub-Saharan Africa (UNICEF, 2018). Within sub-Saharan Africa, the West African region has the highest maternal mortality in Africa, approximately accounting for 20% of global maternal deaths (UNICEF, 2009). In West Africa, Cameroon has one of the highest maternal mortality rates with 596 deaths / 100,000 live births (WHO, 2015). Additionally, in Cameroon, approximately 80,000 women and girls suffer from injuries or disabilities caused by complications during pregnancy and childbirth each year (USAID, 2006). The major causes of maternal mortality in Cameroon are hemorrhage, malaria, complications from unsafe abortion, hypertension, anemia and pneumonia (Tebeu et al, 2015).

The skewed nature of maternal deaths demonstrates that the majority of maternal deaths can be prevented through timely interventions by skilled healthcare providers whether during antenatal, delivery or post-partum period (Kuhnt & Vollmer, 2017). With regards to antenatal care, it serves as a key entry point for a broad range of services that enables the detection and management of risky conditions associated with pregnancy and child birth. The WHO's Antenatal Care guidelines recommend a minimum of eight contacts by pregnant women with skilled healthcare professionals during pregnancy (WHO, 2016).

Early initiation of antenatal care can support micronutrient supplementation and treatment for pregnancy induced hypertension to prevent pre-eclampsia and eclampsia. Access to health care as part of antenatal care can also act as a route to provide immunization against tetanus, HIV testing and medication to prevent Mother -To - Child Transmission (PMTCT) in case of HIV positive pregnant women (Cumber et al, 2016). In Cameroon where malaria is endemic, health personnel also provide pregnant women with medication and insecticide treated mosquito nets during antenatal care visits to help prevent the disease. Moreover, during antenatal care pregnant women are educated on topics like nutrition, medication, lifestyle, exercise, personal and environmental hygiene.

#### 1.2 Problem Statement

Research shows that in Cameroon only about 20.5% of pregnant women initiate antenatal care during the first trimester (Njim, 2016). The North West Region is one of the four regions in the country with estimates of unmet antenatal care needs higher than 40% (Bonono & Ongolo-Zogo, 2012). Late initiation of antenatal care could have negative consequences on overall perinatal outcomes in women and their children hence increasing morbidity and mortality (Njim, 2016). Research on initiation of antenatal care among pregnant women in Cameroon has for the most part centered on determining the proportion of women who initiate antenatal care late during pregnancy (Halle-Ekane et al 2014; Njim, 2016). Most of these studies do not explore in greater depth the perceptions of women to get a deeper understanding of why the majority of pregnant women initiate antenatal care late (Ancbang-Kimbi, 2014; Halle-Ekane et al, 2014; Njim, 2016; Cumber et al, 2016; Tolefac et al, 2017).

## 1.3 Purpose of the Study

This study will explore the perceptions of pregnant women concerning late initiation of antenatal care and how their experiences influence decisions on timing of antenatal care initiation. It will also explore the perceptions of midwives on the reasons for late initiation of antenatal care among pregnant women. The purpose of the study is to identify factors that influence timing of antenatal care initiation among pregnant women to assist planners of health education programmes to develop effective interventions that can empower and influence women to initiate antenatal care early during pregnancy.

## 1.4 Outline of the Study

This study is comprised of six chapters and appendices. Chapter one provides the background to the study by highlighting the significance of early antenatal care, an overview of utilization of antenatal care services in Cameroon, the problem statement and the purpose of the study. Chapter two presents a review of literature related to factors influencing timing of antenatal care initiation and a conceptual framework. The third chapter describes the research methods including participant sampling, methods of data collection and data analysis, strategies to ensure trustworthiness of the study and ethical considerations. Chapter four presents and highlights the findings of the in-depth interviews with pregnant women and midwives. Themes that emerged from the analysis are discussed including direct quotes where necessary. Chapter five further discusses the findings, drawing additional insight by placing them in context of the broader literature. The last chapter provides a summary of the

study and recommendations. The appendices include the information sheet, consent form, interview guide, ethics and administrative approvals.



#### **CHAPTER 2: LITERATURE REVIEW**

#### 2.1 Introduction

This chapter reviews literature relating to initiation of antenatal care and its significance in improving quality of maternal health. The literature reviewed also includes studies that investigated factors associated with initiation of first antenatal care visit and with late initiation of antenatal care in low income countries.

#### 2.2 Antenatal care and maternal health

Antenatal care is a key element of the package of services aimed at improving maternal and child health (Kuhnt & Vollmer, 2017). The low maternal/infant morbidity and mortality rates reported for high income countries compared with extremely high rates in low and middle income countries are attributed to the higher utilization of modern obstetric services by the former (Awusi et al, 2009).

While research has demonstrated the benefits of antenatal care through improved health of mothers and babies, the exact components of antenatal care and what to do at what time have been matters of debate (Lincetto et al, 2006). Initially, the high risk approach aimed to classify pregnant women as low risk and high risk based on predetermined criteria and involved many visits (WHO, 2010). This approach was hard to implement effectively since many pregnant women had at least one risk factor, and not all women developed complications. At the same time, some low risk women did develop complications, particularly during child birth (Lincetto et al, 2006).

After the 2001 systematic review, the World Health Organization (WHO) moved away from the high risk antenatal care model developed largely for high income countries (WHO, 2010). The revised model was based on reduced but goal oriented clinic visits (focused antenatal care) which consisted of at least four visits to a health facility during pregnancy (WHO, 2010). More recently, evidence shows that the focused antenatal care model is probably associated with more perinatal deaths than models that comprise at least eight antenatal care visits (WHO, 2016). Furthermore, evidence suggests that more antenatal care visits, irrespective of the resource setting is probably associated with greater maternal satisfaction than less antenatal care visits (WHO, 2016). Currently, the 2016 WHO antenatal care model states that antenatal care models with a minimum of eight contacts are recommended to reduce perinatal mortality and improve women's experience of care (WHO, 2016).

Research investigating the components of antenatal care has recommended the following services during antenatal care:

- Confirmation of pregnancy;
- Detection of problems complicating pregnancy like anemia, hypertensive disorders, bleeding;
- Response to other reported complaints;
- Tetanus immunization, anemia prevention and control;
- Information and counseling on self-care at home, nutrition, safe sex, breastfeeding, family planning, healthy lifestyle;
- Birth planning, advice on danger signs and emergency preparedness;
- Monitoring of progress of pregnancy and assessment of maternal and fetal wellbeing;
- syphilis testing;
- HIV testing and counseling;
- Intermittent preventive treatment of malaria and promotion of insecticide treated nets (Pell et al, 2013).

Research suggests that in low income countries, particularly sub-Saharan Africa, pregnant women often do not receive the recommended antenatal care services (UNICEF, 2018). To ensure that potential complications are identified in early pregnancy and managed effectively, the WHO recommends that women should initiate antenatal care early and have at least eight contacts with healthcare professionals during pregnancy (WHO, 2016).

#### 2.3 Initiation of first Antenatal Care visit

In West Africa, many pregnant women tend to start antenatal care late especially adolescent women, resulting in them not benefiting from preventive and curative services. In a retrospective study on gestational age at antenatal booking and delivery outcomes in Nigeria, the results reported a prevalence of late booking of 86% (Okunlola et al, 2008). This result is in line with a cross sectional study in southern Benin. In this study in Southern Benin, the primary target was pregnant women attending the antenatal care visit regardless of the length of pregnancy. The results showed that among 301 pregnant women, only 24.6 utilized antenatal care services during the first trimester of pregnancy (Ouendo et al, 2015). Another cross sectional study in The Gambia showed similar results with high rates of late initiation of antenatal care. The study involved 457 women attending six urban and six rural antenatal clinics. The results showed that only 8.1% of the women attended antenatal care within the

first trimester of pregnancy while 62.8% and 29.1% attended their first antenatal care in the second and third trimester respectively (Anya et al, 2008).

Similar to other West African studies, research shows that most pregnant women in Cameroon initiate antenatal care late (Halle- Ekane et al, 2014; Njim, 2016). sectional study in the Muea Health Area in the South West Region of Cameroon that is made up of rural/semi urban settlements, findings show that only 27.2% of the women had their first antenatal care visit in the first trimester (Halle-Ekane et al, 2014). Most of the women (69.1%) had their first visit in the second trimester and 3.7% had their first antenatal care visit in the third trimester (Halle-Ekane et al, 2014). The research showed that rural residence was associated with lower antenatal care attendance. Semi urban women were more likely to initiate antenatal care early or attend four times or more than rural women. Financial constraints were the most significant barrier to early initiation of antenatal care (Halle-Ekane et al, 2014). This is because payment for services is out of pocket both in private and public health facilities with no exemption schemes. Community health insurance schemes are weak and not effectively utilized by the population. This is somewhat similar to the results of a cross study in a Suburban Hospital in Buea in the South West Region of Cameroon in which findings revealed that while 60.5% of the women attended at least four antenatal care visits before delivery, only 20.5% of the women attended antenatal care during the first trimester of their pregnancy (Njim, 2016). In contrast to the study in the Muea Health Area that is a rural/suburban area, for the study in Buea in the Suburban Hospital, socio-demographic and obstetric factors were not found to be associated with attending antenatal care in the first trimester (Njim, 2016).

## 2.4 Factors associated with late initiation of Antenatal Care

Ndidi and Oseremen (2010) in their study in the Niger Delta in Nigeria discovered that the reasons given by pregnant women for late initiation of antenatal care is rooted in misconceptions and lack of knowledge on the importance of antenatal care. The result of this study showed that 65.6% of the study population booked late due to lack of knowledge or misconception of the purpose of and right time to commence antenatal care (Ndidi & Oseremen, 2010). It showed that women believe that it is not beneficial to start antenatal care in the first trimester as antenatal care is regarded as a curative rather than a preventive service (Ndidi & Oseremen, 2010). Pregnant women therefore view whatever symptoms they face in early pregnancy as normal, mild and not serious enough to need a doctor's attention (Ndidi & Oseremen, 2010). In addition, financial constraints and fear of the consequences of making

the pregnancy public were mentioned by some of the women as reasons why they initiated antenatal care late (Ndidi & Oseremen, 2010).

In another study by Gross et al, in South Eastern Tanzania, it was found that women attended antenatal care late because of inability to recognize pregnancy early; poor accessibility to the health facility owing to distance; difficulties in crossing rivers or poor road conditions; illness or other obligations; and negligence and apathy (Gross et al, 2012).

In a study carried out in the rural health district of Hlabisa, KwaZulu-Natal, South Africa, women identified several reasons that prevented them from seeking antenatal care in time. These included accessibility (those could not afford transport cost to health facility), being unsure that one is pregnant, waiting for the fetus to move, and not seeing any benefit of attending antenatal care. Thus they would rather go late in pregnancy and reduce the number of visits to the health facility (Myer & Harrison, 2003).

Simkhada et al (2008) carried out a systematic review of twenty eight papers to identify and analyze the main factors that affected the utilization of antenatal care services in low and middle income countries. They found out that maternal education, husband's education, marital status, availability of services, costs, household income, women's employment, media exposure, history of obstetric complications, cultural beliefs and ideas about pregnancy and parity were identified as some of the factors associated with initiation and utilization of antenatal care services (Simkhada et al, 2008).

# 2.5 Conceptual Framework WESTERN CAPE

The Health Belief Model is one of the mostly used conceptual frameworks for understanding health behavior. The model postulates that health seeking behavior is influenced by a person's perception of a threat posed by a health problem and the value associated with actions aimed at reducing the threat (Polit & Beck, 2012). In the context of this study, the health belief model is used to identify perceptions of seriousness, susceptibility and barriers that might explain why some women do not initiate antenatal care early. It also identifies possible cues to action and modifying variables that might change the behavior of late antenatal booking.

## 2.5.1 Core assumptions of the health belief model

The Health Belief Model is based on three assumptions:

- That a person will take health related action if that person feels that a negative health condition can be avoided;
- That a person will take action if that person has a positive expectation that by taking a recommended action they will avoid a negative health condition;
- That a person takes a health related action if the person believes that she can successfully take the recommended action (Sharma & Romas, 2012).

There is a strong relationship between the above assumptions and the focus of this study which is on factors underpinning initiation of early antenatal care. The framework addresses major components for compliance with recommended health action namely; perceived susceptibility, perceived severity, perceived benefits, perceived barriers, cues to action and self-efficacy (Polit & Beck, 2012).

#### 2.5.2 Perceived susceptibility related to early antenatal booking

This is a person's perception that a health problem is personally relevant or that diagnosis is accurate (Polit & Beck, 2012). A person's perception that a health problem is personally relevant will contribute to taking of required action by the individual. Pregnant women require understanding that although pregnancy is a normal process that may entail complications, they remain at risk of getting complications. The greater the perceived risk, the greater the likelihood of engaging in a behavior to decrease the risk.

# 2.5.3 Perceived severity due to late initiation of antenatal care

Even when one recognizes personal susceptibility, action will not occur unless the individual perceives the severity to be high enough to have serious implications (Polit & Beck, 2012). This is one's own opinion of how a condition is and what its consequences are if left untreated. When a pregnant woman recognizes that she is susceptible to getting a certain condition it does not really motivate her to take necessary action until she appreciates that getting the condition would have serious physical, psychological and social implications on her, her pregnancy or baby.

## 2.5.4 Perceived benefits of early initiation of antenatal care

Perceived benefits are the patient's beliefs that a given treatment will cure the illness or help prevent it (Polit & Beck, 2012). It is this belief that gives them confidence to take action because they are sure of the resulting benefits. Pregnant women are able to weigh the benefits and advantages of early initiation of antenatal care. Antenatal care improves the survival and

health of babies directly by reducing stillbirths and neonatal deaths and indirectly by providing an entry point for health contacts with the woman at a key point in the continuum of care. Good care during pregnancy is important for the health of the mother and development of unborn baby (WHO, 2016).

## 2.5.5 Perceived barriers contributing to late initiation of antenatal care

Perceived barriers can affect people's decision making to take particular action and include the complexity, duration and accessibility of the treatment (Polit & Beck, 2012). It is only when people realize that they have the capacity to deal with these barriers that they would be able to take necessary action. Pregnant women can face barriers preventing them to initiate antenatal care early specific to their context and need support in overcoming those barriers.

#### 2.5.6 Cues to action

Cues to action refer to the motivation to take action (Polit & Beck, 2012). Pregnant women may know that early antenatal care is important but they need a cue to trigger them to seek healthcare. Cues to action include issues like sudden complication early in pregnancy, health education or community outreach prompting pregnant women to seek healthcare.

## 2.5.7 Self-efficacy

Self-efficacy refers to the strength of an individual's beliefs in his own ability to respond to difficult situations and to deal with any associated obstacles. Among the modifying factors that have been identified are personal variables, patient satisfaction and socio demographic factors (Polit & Beck, 2012). This is confidence in one's ability to be motivated to take action. Pregnant women should have confidence and believe that they are capable of initiating antenatal care at less than 12 weeks of gestation so that any complications can be identified and corrected early.

# 2.5.8 Integration of factors associated with late initiation of antenatal care into the conceptual framework

## **❖** Individual Perceptions

- Perceived susceptibility
  - Lack of knowledge about early booking
  - Misconceptions
  - Not being sick
  - Delays in diagnosing pregnancy

- Parity
- Belief that pregnancy is normal
- Previous obstetric complications

## Modifying Factors

## Perceived severity

- o Belief that pregnancy is normal
- o Previous obstetric complications
- o Medical condition of the current pregnancy
- Media exposure

## • Self -efficacy

- Socio-economic factors
- Cultural beliefs
- o Poverty
- o Household decision making to start antenatal care

- o Marital status
- o Education level of the husband

## **\Likelihood** of Action

#### Perceived benefits

- o Healthy pregnancy
- o Positive pregnancy outcome

#### • Perceived barriers

- o Transport to health facilities
- Staff attitudes
- Clinic booking procedures
- Cost of services
- House hold decision making to access health care
- Opportunity cost (ability to excuse oneself from existing workload and responsibilities)

#### Cues to action

- Health education
- Medical condition of the current pregnancy
- Media exposure
- Effective appointment booking system

## 2.6 Summary

In summary, effective utilization of antenatal care services is a key determinant of maternal health. WHO recommends that pregnant women should initiate antenatal care early and have at least eight contacts with healthcare professionals during pregnancy. In sub Saharan Africa especially in the West African region, most pregnant women do not receive the recommended antenatal care services and initiate antenatal care late. In Cameroon, 79.5% of pregnant women initiate antenatal care after the first trimester of pregnancy. Some of the factors associated with late initiation of antenatal care include lack of knowledge about early booking, delays in diagnosing pregnancy, perception that pregnancy is a normal health condition, cultural beliefs, lack of decision making power by women, transport difficulties, cost of services, clinic booking procedures, medical condition of current pregnancy and previous obstetric conditions. The Health Belief Model can be used to develop a conceptual framework to understand health seeking behavior and timing for antenatal care initiation. The conceptual framework developed from the health belief model groups the factors associated with late initiation of antenatal care into major components namely; perceived susceptibility, perceived severity, perceived benefits, perceived barriers, cues to action and self-efficacy.

UNIVERSITY of the WESTERN CAPE

#### **CHAPTER 3: METHODOLOGY**

#### 3.1 Introduction

This chapter provides a description of the methodology used in this research. It starts with the aims and objectives of the study followed by the research design, participant selection process, data collection and analysis procedures. It ends with the description of ethical considerations and their application in the study.

#### 3.2 Research Aim.

The aim of the study was to understand the reasons why pregnant women initiate antenatal care late in Nkwen Baptist Health Center, North West Region, Cameroon.

## 3.3 Objectives of the Study

The objectives of the study were;

- To explore the perceptions of pregnant women concerning timing of antenatal care initiation,
- To explore the experiences of pregnant women and how these experiences influence their decisions on the timing of antenatal care initiation,
- To explore the perceptions of midwives on the reasons for late initiation of antenatal care among pregnant women.

WESTERN CAPE

## 3.4 Study Design

The study is an exploratory study. It is qualitative given that it seeks to gain a deeper understanding of the perceptions, opinions and experiences of pregnant women and midwives on factors influencing early antenatal care initiation during pregnancy. Qualitative research can develop concepts that enable the understanding of social phenomena in a particular setting with emphasis on the meaning, experiences and views of participants (Pope & Mays, 1995). It acknowledges that people have different views about a problem and give meaning to their experiences as it is lived and felt (Baum, 1995). Qualitative research confirms that social interactions are constructed through interaction between people and argues that meaning does not exist on its own (Baum, 1995). Hence the approach enabled the researcher to collect data through in-depth interviews with an interview guide using questions that were broad and open ended to enable exploration in detail depending on the responses provided. The approach also enabled the researcher to explore the reasons and opinions behind participants

responses through asking probing questions such as why, how and what to gain a deeper understanding of the reasons for late initiation of antenatal care among pregnant women. Understanding this issue from an in-depth perspective of pregnant women helped throw more light on a problem that cost the lives of many women in Cameroon. The information generated will assist planners of health education programmes to develop effective interventions for pregnant women to raise awareness among women of reproductive age on the importance of accessing antenatal care services early to improve maternal health outcomes.

## 3.5 Research Setting

The study was conducted in Nkwen Baptist Health Center a semi urban health center located in the Bamenda Health District in the North West Region of Cameroon. Nkwen Baptist Health Center is a faith based outpatient clinic belonging to the Cameroon Baptist Convention Health Services. The health center has a staff of 144 and an average monthly patient attendance of 12,128. Meanwhile the average monthly antenatal care clinic attendance for pregnant women is 358. It cost at least 13,000 fcfa (\$26) to initiate antenatal care although it may cost slightly less in public health facilities. This excludes other cost like transport expenses to the health facility and feeding expenses during clinics. Payment for services is out of pocket both in private and public health facilities with no exemption schemes.

The Bamenda Health District is an urban and semi-urban area. With about 337,036 inhabitants, the district has 17 health areas and covers a total surface area of 560 square kilometres (Egbe et al, 2016). There is one main hospital (Bamenda Regional Hospital) that functions as a referral hospital for 17 public, 12 lay private and 5 mission health facilities.

The Bamenda health district is located in the North West Region of Cameroon. With Bamenda as its capital city, the North West Region is the third most populated region in Cameroon with an estimated population of more than 1.8 million inhabitants. It has an urban growth rate of 7.95% higher than the national average of 5.6%, and a rural growth rate of 1.16% equal to the national rate (Ambagna et al, 2012). Over 80% of the natives depend on agriculture for their livelihood including a strong livestock sector (Yengoh, 2012). The region has a poverty rate of 51%. The population is young with 62% of its residents below the age of 20 years (Ambagna et al, 2012).

## 3.6 Sampling

The study sample comprised of eighteen pregnant women and three key informant midwives. The inclusion criteria was pregnant women who presented for their first antenatal care after twelve weeks of pregnancy. The exclusion criteria were pregnant women who were less than eighteen years of age, and pregnant women who could not express themselves in English. Participants were also selected through purposive sampling. There were asked some key demographic questions including number of weeks of gestation to determine their eligibility for interview. There were age variations in the recruitment of participants to ensure that data on the opinion and experiences of young women as well as old women were captured. Participants were selected in terms of number of gravida. This is because women who are experiencing pregnancy for the first time will have a different perception of antenatal care than those who have experienced it a number of times. Marital status was also considered to ensure that both married and single women were interviewed as marital status may have an influence on timing for antenatal care initiation. A minimum of three respondents were interviewed in each category of variation in respondents as showed in table 1 below.

Key informants were made up of midwives serving at the antenatal care unit. The inclusion criteria was midwives who had been serving in the antenatal clinic for at least two years. Their recruitment was through the head of the antenatal care unit. These midwives were included on the basis that they had been working and interacting with pregnant women and could provide information on their perceptions and views regarding timing of antenatal care initiation.

**Table 1: Study Sample** 

n = 21					
Type of variation/Respondent	No. of respondents				
Age	18-32 years	3			
	33-49 years	3			
Gravida	Gravida 1, Para 0	3			
	Gravida >=2, Para >=1	3			
Marital Status	Single	3			
	Married	3			
Key informants	Midwives (>= 2 years of service)	3			
Total	1	21			

#### 3.7 Data Collection

**Table 2: Data Collection Schedule** 

Month & Year	Activity			
December 2017	Familiarization with clinic staff and clinic processes			
January and February 2018	In-depth interviews conducted with pregnant women, transcription and analysis			
March 2018	In-depth interviews with Key informant midwives, transcription and analysis			
March 2018 and April 2018	Final analysis and write up			

#### 3.7.1 Data Collection Method

Pregnant women who initiate antenatal care after the first trimester were informed about the study by service providers at the antenatal care unit during the process of provision of antenatal care services. Respondents were only informed of the study at the end of their visit at the antenatal care unit to ensure that the study did not interfere with their access to care. They were informed that participation was voluntary and that if they wish to participant, they will be referred to the researcher for interview in a room in the clinic. Those who accepted to participate were given a slip by the service providers to indicate that they were informed of the study and directed to the researcher for interview.

## WESTERN CAPE

Data collection was through in-depth interviews. Interviews were face to face. This method provided a rich form of data as the participant was visible to the interviewer who could pick up on non-verbal cues. Questions were asked from a predetermined interview guide. These guides had short list of questions with probes that helped direct the interview in a fluid, conversational manner in a particular direction. Probing was a vital tool for ensuring the credibility or true value of the data as it allowed for the clarification of interesting and relevant issues raised by the respondent. This provided rich descriptions of the understanding of study participants and provided a comprehensive picture of the phenomenon. Some of the points that were used to develop the interview guide and probes included; socio-demographic characteristics, obstetric history, feelings about current pregnancy, support by partner and family, perception of value of antenatal care, perception of when to initiate, knowledge of pregnancy problems, source of education regarding pregnancy and experience of antenatal care initiation process and antenatal care services.

Data recording was through audio taping. This allowed the interviewer to prepare transcript for analysis, based on a verbatim account of the interview. With data recording, the interviewer was able go back to the recording multiple times as needed to catch things that were missed. Written notes also recorded information as a supplement to the audio recorded data. Data analysis was alongside data collection and stopped once saturation was reached.

## 3.8 Data Management

Each interview was assigned a code and a date to maintain confidentiality. At the end of each interview, audio recordings were transcribed verbatim and by the researcher and analyzed manually using Thematic Coding Analysis. The researcher's diary notes were collated and analyzed at the end of each day to ensure reflexivity. The notes were referred to during the process of transcription and data analysis. Electronic data such as audio tapes recordings and transcriptions were stored on the researcher's computer with security codes to limit access to anybody out of the study. Backups were maintained on an external hard drive and kept in a locked drawer in the researcher's office at the work place. Hard copies of data such as note books, consent forms were also securely locked in the drawer. Only the researcher was in possession of the key to this drawer.

## 3.9 Data Analysis

Data analysis was done manually using Thematic Coding Analysis. Thematic Coding Analysis is an inductive analysis in which categories or codes are allowed to emerge from the data (Robson, 2011). The five phases of TCA done were as follows; familiarization, coding, identifying themes, reviewing and refining, integration and interpretation (Gibbs, 2007).

#### 3.9.1 Phase 1: Familiarization

The researcher familiarized himself with the data by repeatedly listening to the audio recordings of the interviews. The transcripts were also read through multiple times. This allowed the researcher to develop a deep understanding of the data (Savin-Baden & Major, 2012). The researcher noted down key impressions in his researcher diary during the process of listening to the audio recordings and reading the transcripts.

#### **3.9.2 Phase 2: Coding**

Coding involved desegregating textual data into segments, examining the similarities and differences in data and grouping together conceptually similar data. Coding was done by assigning a key topic or theme to a sentence or paragraph that relates to an issue of interest in

the study (Robson, 2011). As the transcripts were coded, a list of codes was developed with each code briefly defined. Each time a passage was found that could not be coded with any of the previous codes it was assigned else a new code was created (Dey, 1993).

#### 3.9.3 Phase 3: Identifying Themes

Themes were developed by examining codes to identify those that can be grouped together (Tarlor-Powel & Renner, 2003). After reflection, a broad descriptor was assigned to them to identify their commonality. As a list of codes emerged, focus was on broader patterns in the data, combining further coded data into proposed themes or creating new themes (Creswell, 2014).

## 3.9.4 Phase 4: Reviewing and Refining Themes

This phase involved two levels. The first level involved reading the collated extracts for each theme and ensuring that they appear to form a coherent pattern. Once themes captured the collective meanings of the coded data (thematic map), the researcher moved to the next level. In the second level, the researcher studied the individual themes in relation to the data and also whether the thematic map correctly reflects the meanings manifested in the overall data (Gibbs, 2007).

## 3.9.5 Phase 5: Integration and interpretation

This involved describing the scope and content of each theme. It also involved developing an argument using the data that responds to the research questions. The researcher continuously reflected on the setting and context to help interpret the phenomena. The researcher also drew on existing research to inform the interpretation and strengthen and support the argument (Gibbs, 2007).

#### **3.10 Rigor**

Rigor or trustworthiness is a means of judging the credibility and dependability of the study (Creswell & Miller, 2000). Triangulation of data sources was through interviewing pregnant women and midwives. Transferability of the study was facilitated by describing the setting, participants, themes and the assumptions that are central to the study in rich detail (Creswell & Miller, 2000). An audit trail which provides clear documentation of all research decisions and activities to increase credibility of the study was maintained (Anney, 2014). Audit trial was ensured by documenting the inquiry process through journaling and memoing, keeping a research log of all activities, developing a data collection chronology, and recording data

analysis procedures clearly. Key points at the end of each interview were summarised and verified with the participants to ensure that the understanding and interpretation of their experiences and perceptions was accurate. Reflexivity ensured validity of the study by establishing a report on the researcher's assumptions, values, and biases that may shape the inquiry (Billups, 2014). A documentation of personal thoughts and feelings was maintained throughout the study process for personal monitoring and reflections.

## 3.11 Ethical Considerations

Participation in the study was voluntary for pregnant women and midwives. Respondents were only informed of the study by a staff at the end of their visit at the antenatal care unit to ensure that the study did not interfere with their access to care. They were each provided with a letter explaining the study, requesting their participation and assuring them of confidentiality (Participant Information Sheet, Appendix 1). Their consent was sought and a consent form was available for them to sign if they were willing to participate in the research. Participation in the research did not inhibit the respondent's access to care. Anonymity of participants was assured by ensuring that questions that revealed the identity of participants were not asked and that the results were not linked to their identity in anyway. There was also the use of pseudonyms in the presentation of findings to ensure anonymity. It was anticipated that the research was going to cause no harm to the research participants. However, a professional counselor of Nkwen Baptist Health Center was available in case any of the pregnant women required emotional support or counseling as a result of the research process. Ethical clearance was obtained from the Biomedical Research Ethics Committee of the University of the Western Cape (UWC) and from the Institutional Review Board (IRB) of the Cameroon Baptist Convention Health Services. There was also administrative clearance from the Director of Health Services of the Cameroon Baptist Convention Health Services authorizing the researcher to have access to the research participants at Nkwen Baptist Health Center.

#### **CHAPTER 4: FINDINGS**

#### 4.1 Introduction

This chapter presents the findings of the in-depth interviews with pregnant women and midwives. It captures their perceptions on why pregnant women initiate antenatal care late. Following the conceptual framework of the study, the findings are reported under the following themes;

## Perceived susceptibility/ perceived severity

- Value of early antenatal care
- Pregnancy as normal health condition
- Ideal booking time
- Obstetric history

## **Perceived barriers**

- Accessibility of antenatal care services
- Cost of initiating antenatal care
- Distance to health facility

## **Cues to action**

## **Self** –**efficacy**

- Pregnancy disclosure
- Support from spouse
- Reaction from parents

The participants and researcher's reflections about the setting and context were incorporated into the findings to give a richer description of perceptions on reasons for late initiation of antenatal care.

Table 3: Participants socio demographic information

Participant Code	Marital Status	Age	Gravida	Parity
P1	Single	22	1	0
P2	Married	27	2	1
P3	Married	35	3	2
P4	Married	28	2	1
P5	Married	27	2	1
P6	Married	24	2	1
P7	Married	21	3	2
P8	Married	22	2	1
P9	Married	25	2	1
P10	Single	21	1	0
P11	Single	20	1	0
P12	Married	23	2	1
P13	Married	27	3	2
P14	Married	18		0
P15	Married	34	3	2
P16	Married	27	4	0
P17	Married	40 FRSITV	4	3
P18	Married	38	4	3
	WES	LEKN CA	FE	1

#### 4.2 Themes

## 4.2.1 Perceived susceptibility/perceived severity of antenatal care

These are perceptions that early initiation of antenatal care is not relevant or that there are no serious health implications of being pregnant that require early initiation of antenatal care. Themes that emerged under these perceptions were; value of early antenatal care, pregnancy as normal health condition, ideal booking time.

## 4.2.1.1 Value of early antenatal care

Some of the pregnant women had the perception that the main purpose of early initiation of antenatal care was to know the state of the baby and since the baby was not fully formed in

the first trimester, early initiation of antenatal care was perceived as a waste of time or waste of money.

I could not come for ANC by one or two months [of being pregnant] because the fetus was not yet formed so that I can do echography and know how the baby was doing. It was so early, so being so early like that it would have just been waste of time. (P1, single, age 22, parity 0)

You need to go for antenatal care when pregnancy is big so that they can check the baby well. It's just that when I hear someone saying they are going for antenatal at two or three months [of pregnancy], I judge that it's because they have money to waste. I cannot just waste money like that. (P7, married, age 21, parity 2)

Some women recognized the importance of early antenatal care but lacked the insights of its purpose for pregnant women and instead had a general understanding that pregnancy required antenatal care at some point.

Early antenatal is good...because I am pregnant, and it [antenatal clinic] is a place where when you are pregnant and preparing to deliver you must appear. Had it been I was not pregnant, I could not be here, so I believe I am in the right place. (P8, married, age 22, parity 1)

## 4.2.1.2 Pregnancy as a normal health condition

Many of the pregnant women considered pregnancy as a normal life event rather than as a condition that requires attention of health personnel. Some of the pregnant women said they waited to seek antenatal care only when they felt unwell.

For me I don't really see it that necessary to come for antenatal care clinic that early at two or three months because first of all am not sick, am just normal, am fine and there's nothing wrong with me. (P17, married, age 40, parity 3)

There was no problem within the first three months, so if there was a problem, that's when I would have rushed and come earlier. (P2, married, age 27, parity 0)

Key informants also said most of the pregnant women who initiated late perceive antenatal care as a curative rather than as a preventive service.

They [pregnant women who initiate antenatal care late] feel antenatal care is curative issue meanwhile that's not the case. Antenatal care is preventive.....this

causes them to wait until they have a health problem before they come for antenatal care. (KI1)

## 4.2.1.3 Ideal booking time

Some women said because the purpose of initiating antenatal care was to diagnose any problems that the baby may be having, the ideal booking time is after the first trimester when the baby must have been properly formed.

I know that it is normal [ideal time] to come for antenatal clinic as from four or five months... at that time you can be able to know better how the baby is fairing. (P13, married, age 27, parity 2)

Many women did not have correct information on the ideal booking time due to misinformation from family members, inadequate health education during clinics.

As I was growing, my mother used to teach me all those things, that when a woman is pregnant she needs to go for clinic as from 4 to 5 months of pregnancy so that the nurses can know if the baby is doing fine. (P2, married, age 27, parity 1)

[The nurses] in the health talk [education] when I came here last time when I was pregnant [for five months] did not tell us that a pregnant woman should start clinic when she is just one, two or three months pregnant. I have not heard this before, it is very new to me. (P12, married, age 23, parity I)

## 4.2.1.4 Obstetric history

Women with positive obstetric history perceived pregnancy and safe delivery as normal experience and did not see the need to initiate antenatal care early.

WESTERN CAPE

[As for previous deliveries] I did not have complications, am always fine. I always come for antenatal care clinic later than this [five months] usually seven months when its almost time for me to give birth. I have always been delivering safely so I have no problem ... I believe is just going to be the same because the previous ones I just delivered safely, and this is even the forth pregnancy. (P17, married, age 40, parity 3)

Key informant midwives said positive obstetric history caused some pregnant women to see antenatal care as a routine and preferred to book later. As women deliver more some of them think they know much and will not want to come and book earlier. They think that antenatal care clinic is just a routine, they just think that since they have been going for antenatal care clinic for the previous pregnancies there's no need booking early. (KI2)

Among the pregnant women, there was one case with a negative obstetric history. Due to her blood group and that of her spouse, all their previous children had the sickle cell disease and did not survive. This influenced her to delay initiating antenatal care because she was contemplating terminating the pregnancy.

I lost two children in the past due to our electrophoresis status [incompatibility of her blood group with that of her spouse]. I aborted the third and this is the forth pregnancy and I am not happy about it all...I decided to come now because I was still thinking whether to keep the pregnancy or not [terminate]. (P16, married, age 27, parity 0)

## 4.2.2 Perceived barriers to antenatal care

These are barriers that prevent pregnant women from initiating antenatal care early. Themes that emerge under perceived barriers are; accessibility of antenatal care services and distance to health facility.

# 4.2.2.1 Accessibility of antenatal care services TY of the

Some women said the booking system was user-unfriendly, with long waiting times and some of the staff were rude making accessibility of services difficult and stressful. This influenced their timing of booking antenatal care.

The problem is the place is too congested, the population is too much, when you come you need to stand on a very long line and aahh its really stressing...standing on the long lines every month from the first month [of pregnancy] and for nine months is something I can't really do..... So I decided to come from five months to the last month so that at least I will not have to stress a lot. (P11, single, age 20, parity 0)

I was not really pleased with the way the welcome was at the clinic, some of them are very rude, they don't take time to explain things and end up just shouting at us and that's even the most reason why some of us don't like to come early for clinic because we don't want to interact with them. (P18, married, age 38, parity 3)

## 4.2.2.2 Cost of initiating antenatal care

Some of the pregnant women said it was expensive to initiate antenatal care. They had to delay initiating care because they needed to plan and raise money to pay for the services.

Let me say within the first two months, things were really difficult for us, so even if I was to start by then, I won't have started. Because you know the town is shaking [socio political tensions] now so everything is difficult. Money is difficult to get ...... there would have been no money to pay for tests and drugs within the first two months. (P5, married, age 27, parity 1).

Some women said they could not afford to pay for antenatal care services and delayed initiating antenatal care in order to reduce the number of clinic visits, thereby reducing the total cost of antenatal care during the entire pregnancy period.

You know there are financial challenges, there is a lot of hardship here and you have to pay for the cost of antenatal care ... to start coming from the first month [of pregnancy] to the last month like that I don't really have money because it is expensive to be coming from the first month to the last month, no, no, I cannot afford money to pay. (P17, married, age 40, parity 3)

Midwives concurred that initiating antenatal care is expensive (\$26) to many pregnant women and lack of finances is one of the reasons why many of them book late. This amount is too high for women within this community to afford in order to initiate antenatal care.

For first booking you spend at least 13,000 fcfa [\$26] and they always see early booking to be expensive to them..... we always at least attend to them and give them services according to the money they are able to have and tell them to go and look for money and come and finish their lab tests. (KI3)

#### 4.2.2.3 Distance to health facility

Some women said that the distance to the health facility was far and that transportation difficulties to get to the facility caused them to postpone initiating antenatal care early.

I do have difficulties of transport to come for clinic. You know the distance is far and I use bike [motor cycle], am always very dizzy, that makes it difficult [to initiate clinic early] (P11, single, age 20, parity 0)

Even though we have tarred road but the only means of transport is bike we don't have taxi. It's difficult with this pregnancy to climb on a bike, you are not confortable, you are not sitting well so most at times you find yourself trekking for long to where you can see a taxi to come for the clinic ...when you just think how you start trekking or climbing on a bike and start rolling down a long distance with all the wind it discourages you from going [initiating) for clinic early. (P18, married, age 38, parity 3)

#### 4.2.3 Cues to action

These are triggers that can cause a pregnant woman to take necessary action to initiate antenatal care early. The absence of these cues can caused pregnant women to initiate antenatal care late. The theme that emerged under cues to action was community health education.

## 4.2.3.1 Community health education

The absence of community outreach programs that could sensitize women on the need to initiate antenatal care early caused some pregnant women to initiate late.

To say health workers come to the community to educate us on how to go about [early antenatal care initiation] when you are pregnant, I have not seen that... [As far as] seeing a doctor or a nurse coming around our quarter to help us enlighten [educate] us on pregnancy and [early] antenatal care, I have never seen. (P18, married, age 38, parity 3)

These pregnant women are ignorant on things about pregnancy and [early] antenatal, they lack education...we lack a forum where we can really educate women in the community on early start of antenatal clinic... (KI1)

#### 4.2.4 Self – Efficacy

This is the confidence that enables a pregnant woman to be motivated to take action. It is influenced by socio economic and demographic factors. In this study, some pregnant women did not believe that they were capable of taking the decision to initiate antenatal care within the first trimester. Themes that emerged under self – efficacy were; cost of initiating antenatal care, pregnancy disclosure, support from spouse, reaction from parents

## 4.2.4.1 Pregnancy disclosure

Some of the participants initiated antenatal care late because they wanted to delay making the pregnancy public because of fear of perceived "enemies" who may harm their pregnancy.

I did not come before this time because I did not want people to know especially people who don't wish me well, my enemies. (P2, married, age 27, parity 1)

Other women said they delayed making their pregnancy public because they were shy or ashamed when the pregnancy was still small. It was noted that stigma associated with early pregnancy disclosure influenced both married and unmarried women on the timing for antenatal care booking.

Pregnancy in our culture even though you are married it has some types of conceptions. At times I am shy and so I will not want my neighbors and people around to first of all know .... Culturally you feel shy.... even though married, it has a little aspect of shame related, you don't feel confortable you just feel a type [unconfortable]. (P18, married, age 38, parity 3).

One of the key informant midwives said unmarried women especially young girls also hide the pregnancy within the first trimester due to the shame that information of their pregnancy will bring on their parents.

Most pregnant women at the beginning of pregnancy are always shy especially those who are not married, they shy away first of all because they don't want their neighbors, or their immediate family members to know that they are pregnant so they hide the pregnancy seriously.....some are ashamed for fear of stigma that their neighbors will laugh at their parents that though she was so holy she is not married but is pregnant. (KII)

Some of the women said community members consider early antenatal care as a show of pride and do mock at women who initiate antenatal care early.

So for us we believe that you only start going for clinic when the stomach is already very big as from six months. Because when you go for antenatal care at one or two months when the baby is still small is like you are boosting of something, proud which does not really speak well of you[in the community]. (P17, married, age 40, parity 3)

## 4.2.1.2 Support from spouse

In some cases, lack of support of the spouse contributed to late initiation. Lack of trust made some husbands not to believe their wives when they (wives) told them that they were pregnant. This made the husbands to be reluctant to provide money for early initiation of antenatal care.

Whenever I tell the father of my children that I am pregnant he usually take it for a lie...each time I request for money to go for clinic he is not willing and will ask me to wait and he will give it [money] at his own time. (P7, married, age 21, parity2)

Lack of knowledge on ideal booking time by husbands also contributed in weakening the support they gave to their wives to initiate antenatal care early

I was not given money on time by my husband and when I said I was pregnant and needed to go for antenatal care early he thought I was lying...It took many months before he gave me money..... He thought one needed to go for antenatal care at 6 months [of pregnancy]. (P14, married, age 18, parity 0)

Marital misunderstanding was also identified as one of the reasons that caused many husbands not to support their wives to book early.

Sometimes he [husband] is not understanding, what I will actually want from him he will not even give me. Like this food they are telling us to go and eat, I don't know how I will explain to him because according to him he thinks that I just want to take his money and eat... he just get angry and say, why are you struggling to go, you just want to waste my money. (P18, married, age 38, parity 3).

## **4.2.1.3 Reaction from parents**

Most of the unmarried women especially young girls said fear of negative reaction from parents led to late disclosure of pregnancy and hence contributed to late initiation of antenatal care.

My parents were not going to welcome the pregnancy since I was just a student ...so telling them when the pregnancy was still one or two months or so it would have been a taboo or something and I will surely be beaten...my parents are wild and there could do anything, so I was scared [and decided to hide the pregnancy from them]. (P11, single, age 20, parity 0)

#### **CHAPTER 5: DISCUSION**

This chapter discusses the findings of the study. It will focus on key findings and place them in the context of the broader literature.

## 5.1 Perceived susceptibility/ Perceived Severity due to late antenatal care initiation

A major finding of the study was the lack of knowledge on the purpose of early antenatal care and therefore the right time to initiate antenatal care. This lack of understanding is also influenced by a perception that antenatal care is primarily to detect or treat diseases. This explains reasons why many participants said they did not have any problems in early pregnancy that needed the intervention of health personnel. Some respondents assumed that there were no benefits in booking in the first three months. There is a perception that women can successfully go through the first trimester of pregnancy without antenatal care. They view whatever health issues as a normal health condition or not serious to require that they seek for healthcare. These are the arguments that were used by those that were advocating for a goal oriented antenatal care visits. Hence antenatal care is perceived as a curative rather than a preventive intervention. This is in line with a study by Ndidi and Oseremen in which they reported that most women book antenatal care late because of the belief that there are no advantages in booking for antenatal care in the first three months of pregnancy (Ndidi & Oseremen, 2010). Some of the women were aware of the importance of early antenatal care but lacked insight into its comprehensive purpose. The value of early initiation of antenatal care was not well described and most often focused on curative or as preparation for delivery, as was found in a study in rural South Africa (Myer & Harrison, 2003).

In this study, some participants believed that there was no ideal booking time for antenatal care similar to a study in Southern Nigeria in which the majority of pregnant women claimed that it was safe to book antenatal care at any time during pregnancy (Utuk et al, 2017). There were diverse reasons for lack of information on ideal booking time. Some participants responded that they were never informed of the ideal booking time by service providers during previous antenatal clinics. Health education programmes during antenatal care clinics failed to address the issue of ideal booking time and multi gravida cases who booked late in previous pregnancies were likely to continue with the same practice during subsequent pregnancies. In a study in Buea Health District in Cameroon, few and ineffective health education sessions by service providers during antenatal care clinics was highlighted to be related to poor utilization of antenatal care services by pregnant women (Halle Ekane et al,

2015). It has been found that past experience on antenatal care service is not a predictor for timely booking of antenatal care (Alemayehu et al, 2010).

In this study, some participants responded that they grow up and observed their mothers initiate antenatal care later in pregnancy. Others said they were advised by their mothers or spouses to initiate antenatal care after the first trimester. The study reveals the important role parents or spouses play in deciding the time of booking for antenatal care. There is need to develop health education programs that empower parents and spouses to improve their knowledge on the importance of early antenatal care services. In a study in southern western Nigeria incorrect advice on the best time to start antenatal care from relatives or partners was highlighted as some of the reasons that women in their first pregnancies were starting antenatal care late (Adekanle & Isawumi, 2008).

Multigravida participants said they used previous positive pregnancy outcomes as experience for them in handling subsequent pregnancies. Previous positive pregnancy experiences made pregnant women develop confidence and thus were less motivated to initiate antenatal care early. This is in line with another study that found that multiparous women are usually confident, believing that having delivered many times previously, they are well versed with the art and need not book for antenatal care early (Utuk et al, 2017). On the other hand, previous negative pregnancy outcomes influenced some of the participants to delay initiating antenatal care because initially they planned to terminate the pregnancy. This is similar to a study that found that some women postpone initiating antenatal care until they are free from a perceived obligation to terminate the pregnancy. This may occur with unplanned pregnancies after the woman may have gone through a bad obstetric history (Haddrill et al, 2014).

## 5.2 Perceived health systems barriers to early antenatal care

Some respondents perceived the booking system as user-unfriendly. They complained of overcrowded conditions; a lot of movements between the consultation room, laboratory, ultrasound sound and pharmacy that are far from each other; long waiting times and rudeness of some staff of the clinic. These experiences undermined the quality of antenatal care offered to pregnant women. Women who perceived poor quality services preferred to delay initiating antenatal care to avoid going through the experience at the early stage of pregnancy. Dissatisfaction with care in health facilities including long waiting times, rude and unfriendly attitudes of healthcare providers have been found to be related to late booking among pregnant women (Mrisho et al, 2009).

In this study, some participants expressed their inability to afford the cost of initiating antenatal care and had to delay booking until they raised the required amount. While some women said the cost of initiating antenatal care (\$26) was expensive, others said the negative economic effects of the socio political tensions in the region have aggravated financial hardship limiting their ability to pay for the cost of booking antenatal care early. Booking for antenatal care require payment for a number laboratory tests and for drugs. In addition, pregnant women have to pay for transport to the health facility. Most of the women in this community are poor. Payment for services is out of pocket and there are no exemption schemes. This system renders many of the women unable to afford for health services. In a low resource setting like Cameroon, financial constraints and distance to the health facility plays a major role in determining the timing of initiation of antenatal care. Distance limits the ability and willingness to seek health care where the road network is poor and the common means of transport is by motorcycles. These reasons are similar to a study conducted in Ethiopia where financial constraints where amongst the commonest reasons for late antenatal care booking (Gulema & Berhane, 2017). Tolefac et al also found out that in Cameroon distance to nearest health facility and transport cost are strong barriers to early initiation of antenatal care among pregnant women (Tolefac et al, 2017). Uncomfortable transport and poor road conditions have also been found to be barriers to utilization of antenatal care by pregnant women (Mathole et al. 2004).

#### **5.3** Cues to action

This study found that there was no community outreach that could serve as cue to action for pregnant women to initiate antenatal care early. The absence of a community health education programme contributed to the lack of knowledge on ideal booking time that led to late initiation of antenatal care by pregnant women. If women are to be encouraged to seek antenatal care early, the purpose and value of early initiation of antenatal care will need to be communicated across the communities in which they live. Other studies have found that public health strategies within communities are required to raise awareness and promote early antenatal care services among pregnant women (Titaley et al, 2010).

UNIVERSITY of the

## 5.4 Self – efficacy

In this study, fear of disclosing pregnancy due to community pressures and beliefs was associated with late initiation of antenatal care. Some participants delayed initiation of antenatal care out of shame while others were afraid of being mocked at by community members for initiating antenatal care too early. Fear of bewitchment was also raised as a

reason for booking antenatal care late by some women. Fear of perceived "enemies" who can harm the woman's pregnancy has been found to contribute to late initiation of antenatal care (Ndidi & Oseremen, 2010). This may also support the findings that social norms like seeking advice from village elders before disclosing pregnancy are still dominant in decision-making process on timing for antenatal care initiation (Roberts et al., 2017). Some participants responded that unplanned pregnancies especially among young singles were in most cases associated with late disclosure to the parents for fear of potentially negative reaction. Perceived lack of parental support translated into late initiation of antenatal care. Social support has been shown to facilitate early antenatal care attendance (Abrahams et al, 2001). Lack of support from spouses by not providing the money required to cover the cost of antenatal services or by discouraging early initiation was highlighted by some of the participants as reasons for delayed initiation of antenatal care. These women had to wait for the spouse to decide for them on when to start clinic. The spouse did not either provide the cash to cover the cost of antenatal care or was ignorant of the importance of early antenatal care. In our society, husbands play a key role in decision making for women hence the need to involve men in health education programmes that aim at promoting effective utilization of antenatal care services. Having a spouse who is not supportive was highlighted as being associated with initiating antenatal care late for both adolescent and adult pregnant women in South Eastern Tanzania (Gross el 2012).

## **5.5 Study Limitations**

The respondents are pregnant women who attended antenatal care at Nkwen Baptist Health center while key informant midwives are service providers at this health center. The findings did not capture perceptions of potential respondents who did not visit this particular health facility. It did not also capture perceptions of pregnant women who did not attend antenatal care during pregnancy. These may be women who are most disadvantaged or marginalized in the community. Hence the findings of the study cannot be generalized to them.

UNIVERSITY of the

#### **CHAPTER 6: CONCLUSION AND RECOMMENDATIONS**

#### **6.1 Conclusion**

The study explores the perceptions of pregnant women on reasons why they initiate antenatal care late. In Cameroon only about 20.5% of pregnant women initiate antenatal care within the first trimester of pregnancy. The study revealed the following perceptions of pregnant women as reasons for late initiation of antenatal care.

#### Perceived susceptibility/perceived severity of antenatal care

Pregnant women place low value on the purpose of early antenatal care due to lack of knowledge of its importance. They perceive pregnancy as a normal health condition or not a serious issue that requires seeking health care. Pregnant women lack information on the ideal booking time due to the ineffectiveness of health education programmes during antenatal care clinics. Misinformation from family members and spouses are also identified as reasons for lack of information on ideal booking time for antenatal care by pregnant women.

#### Perceived barriers to antenatal care

Some participants perceive the booking system as user-unfriendly and complained of overcrowded conditions, long waiting times and rudeness of some clinic staff. Women who perceive poor quality services prefer to delay initiating antenatal care in order to avoid going through the experience at the early stage of pregnancy. Cost of initiating antenatal care as well as distance to health facilities with uncomfortable transport and poor road network are also identified as barriers to early initiation of antenatal care.

#### Cues to action

The absence of community health education programmes that could serve as triggers for early antenatal care contributed to the lack of knowledge on ideal booking time. This caused some pregnant women to initiate antenatal care late.

WESTERN CAPE

## Self - efficacy.

Perceived lack of support from parents for unmarried young women and lack of support from spouses for married women translated into late initiation of antenatal care. Fear of bewitchment and stigma due to cultural values and beliefs about early initiation of antenatal care were also identified as variables influencing late initiation.

#### **6.2 Recommendations**

• Community awareness campaigns should be organized and messages should be tailored to address community pressures and stigma that serve as barriers to early

- antenatal care. Messages should also address family decision making dynamics and target decision makers such as parents and husbands and empower women and enable them to be able to make informed decisions regarding their health.
- Other form of media like television, radio, school curriculum and information, education and communication (IEC) materials should be used to raise awareness in the community on the importance of early antenatal care
- Forums should be organized for men in the communities to encourage partner involvement in early antenatal care in particular, but also to enable them support their wives during pregnancy in general.
- Content of health talks in the facility should be reviewed and improved upon to capture accurate information on ideal booking time
- Antenatal care should be done on a daily basis to reduce overcrowding and long waiting times
- Training on interpersonal communication skills should be organized for nurses to improve their attitude and care towards pregnant women
- Create supportive environment at the antenatal care clinic for pregnant women to express their concerns and feelings to enhance client satisfaction
- Government should implement exemption policy for cost of initiating antenatal care

## **6.3 Future Research**

More research is needed to explore the perceptions of pregnant women in different settings to establish whether the findings can be generalized. Also further research is needed to explore the perceptions pregnant women who may not come to a health facility for antenatal care during the entire period of pregnancy.

#### 7.0 REFERENCE LIST

Abrahams, N., Jewkes, R. & Mvo, Z. (2001). Health care-seeking practices of pregnant women and the role of the midwife in Cape Town, South Africa. *J Midwifery Women's Health*. 46(4): 240-247.

Adekanle, D. & Isawuni, A. (2008). Late Antenatal Care Booking and Its Predictors among Pregnant Women in South Western Nigeria. *Online Journal of Health and Applied Sciences*. 7(1):4.

Alemayehu, T., Yilma, M. & Zewditu, K. (2010). Previous utilization of service does not improve timely booking in antenatal care: Cross sectional study of timing of antenatal care booking at public health facilities in Addis Ababa. *Ethiopian Journal of Health Development*. 24: 226-233.

Ambagna, J., Kane, Gilles Q. & Oyekale, S. (2012). Subsistence Farming and Food Security in Cameroon: A Macroeconomic Approach. *Life Science Journal*. 9(4):3949-3954

Anchang-Kimbi, J., Achidi, E., Apinjoh, T., Mugri, R., Chi, H., Tata, R., Nkegoum, B., Mendimi, J., Sverremark-Ekstrom, E. & Troye-Blomberg, M. (2014). Antenatal care visit attendance, intermittent preventive treatment during pregnancy (IPTp) and malaria parasitaemia at delivery. *Malaria Journal*. 13:162

WESTERN CAPE

Anney, V. (2014). Ensuring the Quality of the Findings of Qualitative Research: Looking at the Trustworthiness Criteria. *Journal of Emerging Trends in Educational Research and Policy Studies*. (5(2): 272-281. (Online). Available: <a href="https://pdfs.semanticscholar.org/1419/f7b54e6b7f1215717a5056e0709f8946745b.pdf">https://pdfs.semanticscholar.org/1419/f7b54e6b7f1215717a5056e0709f8946745b.pdf</a>. (27/4/2017 9:02 AM).

Anya, S., Hydara, A. & Jaiteh, L. (2008). Antenatal care in The Gambia: Missed opportunity for information, education and communication. BMC Pregnancy and Childbirth. (8(9).

Awusi, V., Anyanwu, E., Okeleke, V. (2009). Determinants of antenatal care services utilization in Emevor Village, Nigeria. Benin Journal of Postgraduate Medicine. 11(1).

Baum, F. (1995). Researching Public Health: Behind the Qualitative-Quantitative Methodological Debate. *Social Science and Medicine*, 40(4): 459-468.

Billups, F. (2014). The Quest for Rigour in Qualitative Studies: Strategies for Institutional Researchers. (Online). Avaialable: <a href="https://www.airweb.org/eAIR/specialfeatures/Documents/ArticleFBillups.pdf">https://www.airweb.org/eAIR/specialfeatures/Documents/ArticleFBillups.pdf</a>. (27/4/2017 11:05 AM).

Bonono, R. & Ongolo-Zogo, P. (2012). Optimizing the use of antenatal care services in Cameroon. Centre for Development of Best Practices in Health-Central Hospital, Yaounde, Cameroon. (Online). Available: <a href="http://www.who.int/evidence/sure/FRPBCPNEN.pdf">http://www.who.int/evidence/sure/FRPBCPNEN.pdf</a>. (09/10/2016 9:17 PM).

Campbell, O. & Graham, W. (2006). Strategies for reducing maternal mortality: getting on with what works. *The Lancet* 368: 1284-1299.

Carroli, G., Roodney, C. & Villar, J. (2001). How effective is antenatal care in preventing maternal mortality and serious morbidity? An overview of the evidence. *Paediatric and perinatal Epidemiology* 15: 1-42.

Creswell, J. & Miller, D. (2000). Determining Validity in Qualitative Inquiry. *Theory into Practice*, 39(3): 122-130.

RIN RIN RIN

Creswell, J. (2014). Research Design. Qualitative, Quantitative, and Mixed Methods Approaches. Sage Publications.

Cumber, S., Diale, D., Stanly, E. & Monju, N. (2016). Importance of antenatal care services to pregnant women at the Buea Regional Hospital Cameroon. *Journal of Family Medicine and Health Care*. 2(4): 23-29. (Online). Available: <a href="http://article.sciencepublishinggroup.com/pdf/10.11648.j.jfmhc.20160204.11.pdf">http://article.sciencepublishinggroup.com/pdf/10.11648.j.jfmhc.20160204.11.pdf</a>. (27/4/2016 9:32 AM).

Dey, I. (1993). *Qualitative Data Analysis: A User-Friendly Guide for Social Scientists*. London: Routledge.

Egbe, T., Ncham, E., Takang, W., Egbe, E. & Halle Ekane, G. (2016). Use of the Partogram in the Bamenda Health District, North West Region, Cameroon. A Cross-Sectional Study. *Openventio Publishers*, 2(5).

Gibbs, G. (2007). Analyzing Qualitative Data. London: Sage Publications.

Gross, K., Alba, S., Glass, T., Schellenberg, J., Obrist, B. (2012). Timing of antenatal care for adolescent and adult pregnant women in south-eastern Tanzania. *BMC Pregnancy and Childbirth* 12(16): 1-2.

Gulema, H. & Berhane, Y. (2017). Timing of first antenatal care visit and its associated factors among pregnant women attending public health facilities in Addis Ababa, Ethiopia. *Ethiopian Journal of Health Sciences*. 27(2): 139-146.

Haddrill, R., Jones, G., Mitchell, C. & Anumba, D. (2014). Understanding delayed access to antenatal care: a qualitative interview guide. *MBC Pregnancy and Childbirth*. *14*(207).

Halle-Ekane, G., Obinchemti, T., Nzang, N., Mokube, N., Njie, M., Njamen, T. & Nasah, B. (2014). Assessment of the Content and Utilization of Antenatal Care Services in a Rural Community in Cameroon: A Cross-Sectional Study. *Open Journal of Obstetrics and Gynecology*. 4: 846-856.

Halle-Ekane, G., Obinchemti, T., Tamufor, E., Njie, M., Njamen, T. & Achidi, E. (2015). Perceptions of antenatal care services by pregnant women attending government health centers in the Buea Health Districts, Cameroon: a cross sectional study. *Pan African Medical Journal* 21(45): 4858.

## UNIVERSITY of the

Kuhnt, J. & Vollmer, S. (2017). Antenatal care services and its implications for vital and health outcomes of children: evidence from 193 surveys in 69 low-income and middle-income countries. *BMJ Open 7: 17122*.

Lincetto, O., Mothebesoane-Anoh, S., Gomez, P. & Munjanja, S. (2006). Antenatal Care. In: Lawnj., Kerber., eds. Opportunities for Africa's Newborns: Practical Data, Policy and Programmatic Support for newborn and Child Health. (Online). Available: http://www.who.int/pmnch/media/publications/oanfullreport.pdf. (7/10/2016 10:40).

Mathole, T., Lindmark, G., Majoko, F. & Ahlberg, B. (2004). A qualitative study of women's perspectives of antenatal care in a rural area of Zimbabwe. *Midwifery*. 20(2). 118:125.

Mrisho, M., Obrist, B., Schellenberg, J., Haws, R., Mushi, A., Mshinda, H., Tanner, M. & Shellenberg, D. (2009). The use of antenatal and postnatal care: perspectives and experiences

of women and health care providers in rural southern Tanzania. BMC Pregnancy and Childbirth. 9:10.

Myer, L., Harrison, A. (2003). Why do women seek antenatal care late? Perspectives from rural South Africa. *Journal of Midwifery and Women's Health.* 48 (4): 268-272.

Ndidi, E., Oseremen, I. (2010). Reasons given by pregnant women for late initiation of antenatal care in the Niger Delta, Nigeria. *Ghana Medical Journal*. 44(2): 47-51.

Njim, M. (2016). Late Pregnancy Outcomes among Women who attended and Women who did not attend First Trimester Antenatal Care Visits in a Suburban Regional Hospital in Cameroon. *International Journal of MCH and AIDS (2016)*. 4(1): 14-23.

Okunlola, M., Owonikoko, K. & Fawole, A. (2008). Gestational age at antenatal booking and delivery outcome. *Afr J Med Med Sci.* 37(2):165 -169.

Ouendo, E., Sossa, J., Saizonou, J., Guedegbe, C., Mongbo, A., Mayaki, A. & Ouedraogo T. (2015). Determinants of Low Antenatal Care Services Utilization during the First Trimester of Pregnancy in Southern Benin Rural Setting. *Universal Journal of Public Health*. 3(5): 220-228.

Pell, C., Menaca, A., Were, F., Afrah, N., Chatio, S. (2013). Factors Affecting Antenatal Care Attendance: Results from Qualitative Studies in Ghana, Kenya and Malawi. *PLoS ONE* 8(1).

## WESTERN CAPE

Polit, D. & Beck, C. (2012). *Nursing Research, generating and assessing evidence for nursing practice*. China: Wolters Kluwer Health/Lippincott/Williams & Wilkins.

Pope, N. & Mays, C. (1995). Reaching the parts other methods cannot reach: an introduction to qualitative methods in health and health services research. *British Medical Journal*, 311 (6996):42-45.

Roberts, J., Hopp, M., Sealy, D., Taylor, M., Mataya, R. & Gleason, P. (2017). The Role of Cultural Beliefs in Accessing Antenatal Care in Malawi: A Qualitative Study. *Public Health Nursing*. *34*(1): 42-49.

Robson, C. and McCartan, K. (2011). *Real World Research*. West Sussex: John Wiley and Sons.

Savin-Baden, M. & Major, C. (2012). Qualitative Research: *The Essential Guide to Theory and Practice*. Routledge.

Sharma, M. & Romas J. (2012). *Theoretical foundations of health education and health promotion*. Sudbury MA: Jones and Bartlett.

Simkhada, B., Teijlingen, E., Porter, M., Simkhada, P. (2008). Factors affecting utilisation of antenatal care in developing countries: Systematic review of the literature. *Journal of Advanced Nursing*. 61 (3): 244-260.

Tarlor-Powel, E. & Renner, M. (2003). Analyzing Qualitative Data. (Online). Available: <a href="http://www.iisgcp.org/pdf/glssn/Supplemental\_Reading\_on\_Coding\_1.pdf">http://www.iisgcp.org/pdf/glssn/Supplemental\_Reading\_on\_Coding\_1.pdf</a>. (11/06/2017 7:22AM).

Tebeu, P., Halle-Ekane, G., Itambi, M., Mbu, R., Mawamba, Y. & Fomulu, J. (2015). Maternal Mortality in Cameroon: a University Teaching Hospital Report. *The Pan African Medical Journal*. 21(16). (Online). Available: <a href="https://www.ncbi.nlm.nih.gov/pmc/articles/PMC4561158/">https://www.ncbi.nlm.nih.gov/pmc/articles/PMC4561158/</a>. (29/4/2017 8:34 AM).

Titaley, C., Hunter, C., Heywood, P. & Dibley, M. (2010). Why don't some women attend antenatal and postnatal care services?: a qualitative study of community members' perspectives in Garut, Sukabumi and Ciamis districts of West java Province, Indonesia. *BMC Pregnancy and* Childbirth. 10(61): 1471-2393.

WESTERN CAPE

Tolefac, P., Halle-Ekane, G., Agbor, V., Sama, C., Ngwasiri, C. & Tebeu, P. (2017). Why do pregnant women present late for their first antenatal consultation in Cameroon? *Maternal Health, Neonatology, and Perinatology.* 67(8).

United Nations Children Fund (UNICEF). (2009). Maternal and Newborn health – West and Central Africa. (Online). Available: <a href="https://www.unicef.org/wcaro/overview\_2637.html">https://www.unicef.org/wcaro/overview\_2637.html</a>. (20/8/2017 9:54 AM)

United Nations Children Fund (UNICEF). (2018). UNICEF Data: Monitoring the Situation of Children and Women. (Online). Available: http://data.unicef.org/topic/maternal-health/maternal-mortality/. (10/10/2016 6:33 PM)

United Nations Children Fund (UNICEF). (2018). UNICEF Data: Monitoring the Situation of Children and Women. (Online). Available: https://data.unicef.org/topic/maternal-health/antenatal-care/. (14/07/2017 9:00 PM)

United States Agency for International Development (USAID). (2006). Maternal and Neonatal Program Effort. (Online). Available: <a href="http://www.policyproject.com/pubs/MNPI/MNPI2005/2005Cameroon.pdf">http://www.policyproject.com/pubs/MNPI/MNPI2005/2005Cameroon.pdf</a>. (10/10/2016) 5:40 PM).

Utuk, N., Ekanem, A. & Abasiattai, A. (2017). Timing and reasons for antenatal care booking among women in a tertiary health care center in Southern Nigeria. *International Journal of Reproduction, Contraception, Obstetrics and Gynecology.* 6(9):3731-3736.

World Health Organization (WHO). (2010). Integrated Management of Pregnancy and Childbirth – WHO Recommended Interventions for improving Maternal and Newborn Health. Geneva: World Health Organization.

World Health Organization (WHO). (2002). Antenatal Care randomised trial: Manual of the Implementation of the new model. Geneva: *World Health Organization*.

World Health organization (WHO). (2015). Trends in maternal mortality 1990 to 2015. (Online). Available: <a href="http://www.afro.who.int/sites/default/files/2017-05/trends-in-maternal-mortality-1990-to-2015.pdf">http://www.afro.who.int/sites/default/files/2017-05/trends-in-maternal-mortality-1990-to-2015.pdf</a>. (20/8/2017 10:17 AM).

World Health Organization (WHO). (2016). WHO recommendations on antenatal care for positive pregnant experience. (Online). Available: <a href="http://www.who.int/reproductivehealth/publications/maternal\_perinatal\_health/anc-positive-pregnancy-experience/en/">http://www.who.int/reproductivehealth/publications/maternal\_perinatal\_health/anc-positive-pregnancy-experience/en/</a>. (26/08/2017 10:34).

World Health Organization (WHO). (2018). Maternal Mortality. (Online). Available: http://www.who.int/news-room/fact-sheets/detail/maternal-mortality. (13/07/2017 5:35PM).

Yengoh, G. (2012). Determinants of yield differences in small-scale food crop farming systems in Cameroon. (Online). Available: <a href="https://agricultureandfoodsecurity.biomedcentral.com/articles/10.1186/2048-7010-1-19">https://agricultureandfoodsecurity.biomedcentral.com/articles/10.1186/2048-7010-1-19</a>. (27/08/2017 12:21 PM).

#### 8.0 APPENDICES

## **Appendix 1. Participant Information Sheet**



## University of the Western Cape

Private Bag X 17, Bellville 7535, South Africa

Tel: +27 21-959 2809 Fax: 27 21-959 2872

E-mail: soph-comm@uwc.ac.za

## **INFORMATION SHEET**

**Project Title:** Perceptions of pregnant women on reasons for late initiation of antenatal care in Nkwen Baptist Health Centre, North West Region, Cameroon.

UNIVERSITY of the

#### What is this study about?

This is a research project being conducted by Denis Warri at the University of the Western Cape in South Africa. We are inviting you to participate in this research project because you are pregnant and have started your antenatal care after 12 weeks of pregnancy. The purpose of this research project is to identify factors that influence timing of seeking antenatal care among pregnant women. This is intended to gather information that can assist planners of health education programmes to develop effective interventions that can empower and influence women to start antenatal care early during pregnancy. In this way, the health status of women will be improved.

## What will I be asked to do if I agree to participate?

You will be asked to give responses to questions related to why you decided to come for antenatal clinic at this time of your pregnancy. You will also be asked to explain some experiences that you may have gone through that have influenced your decision to start antenatal care at this time of your pregnancy. The interview will be take place in a room at the Antenatal care clinic and will take between 30 minutes to one hour of your time.

## Would my participation in this study be kept confidential?

The researcher undertakes to protect your identity and the nature of your contribution. To ensure your confidentiality, all records of your participation will have locked filling cabinets and storage areas, using identification codes only on data forms, and using password-protected computer files. If we write a report or article about this research project, your identity will be protected.

#### What are the risks of this research?

There may be some risks from participating in this research study. All human interactions and talking about self or others carry some amount of risks. We will nevertheless minimize such risks and act promptly to assist you if you experience any discomfort, psychological or otherwise during the process of your participation in this study. Where necessary, an appropriate referral will be made to a suitable professional for further assistance or intervention. The interview will not in any way interfere with your ability to seek and receive health care.

#### What are the benefits of this research?

This research is not designed to help you personally, but the results may help the investigator learn more about reasons for late initiation of antenatal care by pregnant women. We hope that, in the future, other people might benefit from this study through improved understanding of the reasons why pregnant women initiate antenatal care late in pregnancy.

#### Do I have to be in this research and may I stop participating at any time?

WESTERN

Your participation in this research is completely voluntary. You may choose not to take part at all. If you decide to participate in this research, you may stop participating at any time. If you decide not to participate in this study or if you stop participating at any time, you will not be penalized or lose any benefits to which you otherwise qualify.

#### What if I have questions?

This research is being conducted by Denis Warri of the School of Public Health at the University of the Western Cape. If you have any questions about the research study itself, please contact Denis Warri at: Directorate of Health Services, Baptist Center, Nkwen Bamenda, Tel: +237675795010, email: deniswarri@yahoo.com

Should you have any questions regarding this study and your rights as a research participant or if you wish to report any problems you have experienced related to the study, please contact:

Prof Uta Lehmann

School of Public Health

Head of Department

University of the Western Cape

Private Bag X17

Bellville 7535

soph-comm@uwc.ac.za

Prof Anthea Rhode

Dean of the Faculty of Community and Health Sciences

University of the Western Cape

Private Bag X17

Bellville 7535

chs-deansoffice@uwc.ac.za

BIOMEDICAL RESEARCH ETHICS ADMINISTRATION

Research Office

New Arts Building,

C-Block, Top Floor, Room 28

University of the Western Cape

Private Bag X17

Bellville 7535

Tel: 021 959 2988

Email: research-ethics@uwc.ac.za



## **Appendix 2. Informed Consent Form**



# University of the Western Cape

Private Bag X 17, Bellville 7535, South Africa

Tel: +27 21-959 2809, Fax: 27 21-959 2872

E-mail: soph-comm@uwc.ac.za

#### CONSENT FORM

**Title of Research Project:** 

Perceptions of pregnant women on reasons for late initiation of antenatal care in Nkwen Baptist Health Centre, North West Region, Cameroon

The study has been described to me in the language that I understand. My questions about the study have been answered. I understand what my participation will involve and that my responses will only be audio-taped if I agree to that and I agree to participate of my own choice and free will. I understand that my identity will not be disclosed to anyone. I understand that I may withdraw from the study at any time without giving a reason and without fear of negative consequences or loss of benefits.

Participant's name
Participant's signature
Date

## **Appendix 3. Interview Guide**

#### **Interview Guide – Pregnant Women**

1. Please can you tell me why you come for antenatal care clinic?

## **Prompts**

- a. What do you know about antenatal care for pregnant women?
- b. What do you think are the benefits of attending antenatal care clinic?
- c. How did you get information on the importance of antenatal care clinic?
- 2. Share with me some of the reasons why you decided to start antenatal care after three months of your pregnancy?

#### **Prompts**

- a. How do you feel being pregnant? Why?
- b. What are some of the difficulties you have faced since you became pregnant?
- c. For multi gravida cases what are some of the difficulties you faced with your previous pregnancy?
- d. How do you feel with the way nurses and midwives here attend to you when you come to this health center?
- 3. Whose advice do you take in your family and community with regards to antenatal care?

## **Prompts**

- a. What are some of the things they tell you about antenatal care?
- b. Why do you trust or follow their advice?
- c. How do you feel when people know you are pregnant when you are just one or two months pregnant? Why?
- 4. Can you describe some of the things you go through or experience during pregnancy?

#### **Prompts**

- a. How are you treated by your spouse and family members during this period you are pregnant?
- b. How do you get money to take care of yourself during pregnancy?
- c. How do you get money to prepare for the baby's needs?

- d. For multi gravida cases what are some of the experiences you you had with your previous pregnancy?
- e. What difficulties do you have travelling from your home to attend antenatal care clinic?
- 5. Please share with me how some of these experiences may influence your decision to start antenatal care after three months of pregnancy.

## **Prompts**

a. Which of these experiences caused you to decide to come to antenatal care after three months of pregnancy?

## **Interview Guide – Key Informants**

1. Please can you share with me why most women start antenatal care after three months of pregnancy?

## **Prompts**

- a. How does this vary by age, gravida/parity or marital status?
- b. What perceptions do pregnant women have concerning starting antenatal care within the first three months of pregnancy?
- 2. Please share with me some of the things or experiences pregnant women go through that may influence their decision to initiate antenatal care late

#### **Prompts**

- a. How do people in the community feel about a woman who starts antenatal care within the first one to three months?
- b. What are some of the things you think pregnant women feel uncomfortable with in the clinic when they come for antenatal care?

WESTERN CAPE

# **Appendix 4. UWC Ethics Approval**





# OFFICE OF THE DIRECTOR: RESEARCH RESEARCH AND INNOVATION DIVISION

Private Bag X17, Bellville 7535 South Africa T: +27 21 959 2988/2948 F: +27 21 959 3170 E: research-ethics@uwc.ac.za www.uwc.ac.za

14 December 2017

Mr D Warri School of Public Health Faculty of Community and Health Sciences

Ethics Reference Number: BM17/10/20

Project Title:

Perceptions of pregnant women on reasons for late initiation of antenatal care in Nkwen Baptist Health

Center, North West Region, Cameroon.

**Approval Period:** 

14 December 2017 – 14 December 2018

I hereby certify that the Biomedical Science Research Ethics Committee of the University of the Western Cape approved the scientific methodology and ethics of the above mentioned research project.

Any amendments, extension or other modifications to the protocol must be submitted to the Ethics Committee for approval.

Please remember to submit a progress report in good time for annual renewal.

The permission from the Health Department/Facility must be submitted for record keeping purposes.

The Committee must be informed of any serious adverse event and/or termination of the study.

pres

Ms Patricia Josias Research Ethics Committee Officer University of the Western Cape

PROVISIONAL REC NUMBER -130416-050

FROM HOPE TO ACTION THROUGH KNOWLEDGE





#### CAMEROON BAPTIST CONVENTION HEALTH SERVICES

Director of Health Services P.O. BOX 1 Bamenda, North West Region Republic of Cameroon TEL: (+237) 677 764 781

Email: piustih@cbchealthservices.org
Website: www.cbchealthservices.org

Ref. CBC/DHS-L/17/5827 December 5, 2017

Mr. Denis Warri Chief of Administration and Finance Cameroon Baptist Convention Health Services

Dear Mr. Warri,

## RE: PERMISSION TO CARRY OUT RESEARCH AT NKWEN BAPTIST HEALTH CENTER

I have received your letter dated December 4, 2017 in which you are applying for permission to interview pregnant women at Nkwen Baptist Health Center as part of your research project titled "Perceptions of pregnant women on reasons for late initiation of antenatal care in Nkwen Baptist Health Center, North West Region, Cameroon".

In response, I wish to inform you that your application has been approved. Once you shall receive ethical clearance from the Institutional Review Board (IRB) of the Cameroon Baptist Convention Health Services, you shall be free to contact the Administrator of Nkwen Baptist Health Center to enable you have access to the pregnant women for data collection.

JNIVERSITY of the

Sincerely,

STERN CAPE

Prof Tih Pius Muffih, MPH, PhD Director of Health Services

Cc: The Administrator, Nkwen Baptist Health Center

Appendix 6. CBC Health Board Institutional Review Board Approval Letter

# CAMEROON BAPTIST CONVENTION HEALTH BOARD INSTITUTIONAL REVIEW BOARD

Baptist Centre, Nkwen, P.O. Box 1, Bamenda, Northwest Region

December 19, 2017

Mr. Warri Denis, deniswarri@yahoo.com

Re: IRB2017-30, "Perceptions of pregnant women on reasons for late initiation of antenatal care in Nkwen Baptist Health Centre, North West Region, Cameroon"

Dear Mr. Warri.

Your proposed research will explore the perceptions of pregnant women concerning late initiation of antenatal care and how their experiences influence decisions on timing of antenatal care initiation.

Your study protocol was reviewed by two members of the CBC Health Board IRB and was presented to the entire Board on 10 November 2017.

Please understand that this is the ethical and safety approval for your study. You must present this IRB approval letter to the Assistant Administrator of Nkwen Baptist Health Center for approval to do the study in that institution.

If you make any changes in the research protocol, please immediately send the IRB an amendment specifying the changes proposed.

The Board grants approval for this study for a one-year time period. Thereafter, before December 19, 2018, you will please complete our renewal form/final report which will be attached to an email and returned to the CBC IRB secretariat. The completed form must be reviewed and approved by the Institutional Review Board prior to the expiration date of the current approval period. The fee to renew a study protocol is 10,000 cfa.

Your protocol has been assigned the above reference IRB protocol number. All correspondence to us should include

1) The IRB protocol number

2) Name of the principal investigator and

3) Full title of the study.

Finally, all abstracts, manuscripts, posters and presentations pertaining to the above protocol, must be submitted to the IRB for pre-publication approval.

Please feel free to contact me with any questions and/or concerns regarding the above. Copies of all correspondence regarding this proposal should be sent to me and to Zita Acha secretary, e-mail CBCHBIRB@gmail.com.

Sincerely.

Samuel Ngum Interim Chairperson

> Nancy Palmer, Ph.D., Chairperson, <u>palmernancylea@gmail.com</u> Mrs. Acha Zita, Secretary, <u>cbchbirb@gmail.com</u>